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Remittance Address

Transcare
PO Box 933493
Atlanta, GA 31193-3493

soice Number: 10-07310

I To:

HARRIS COUNTY TOLL RD AUTHORITY
ACCOUNTS PAYABLE

330 MEADOWFERN, SUITE 200
HOUSTON, TX 77067

1stomer Number: HO013

1stomer PO Number: P149763

oject Number: 110000

ojuit Name: Maint VES Project 1300152
FFmS: NET 30

e Date: 12/30/2010

i VES Maint Techs
LL SCHEDULE ITEMS

voice Total

Billing Period From: 11/01/2010
To: 11/30/2010

Current
Amount

58,333.33

58,333.33

$8,333.33
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HCTRA Monthly VES Maintenance Report - NOV10

"j

SNIEFLABNGRERE Date LA [RIRA T3 | EEnLE Lio catio n IAERSIGS | FEZR La ne I8 | iR B L A R DI aw S SR RIHITEY Servicel Performed & M R E AT TG G0 oy pid@ai= iyl fOriginal IEquipment installiDate
Monday, November 01, 2010 ALL ALL Reviewed ATTLAS image report B
Monday, November 01, 2010 ALL ALL Completed moming image checks of all lanes
Monday, November 01, 2010 ALL ALL Continued review of bad OCR data o
Monday, November 01, 2010 ALL ALL Worked on bench testing removed VES compulers/LEDs and preparing them to retum to sarvice
Monday, November 01, 2010 Airport Connector ALL Intermittant comm Issue on some of the lanes, went and swapped octopus cable and confirmed proper
operalion
Monday, November 01, 2010 NEBW8 Work John Ralston__|VESC configuralion/instaliation
Monday, November 01, 2010 ALL ALL Completed aftemoon image checks of all lanes
Tuesday, November 02, 2010 ALL ALL Reviewed ATTLAS image report
Tuesday, November 02, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 02, 2010 ALL ALL Continued review of bad OCR data
Tuesday, November 02, 2010 Fallbrook Exit ALL Video quality issue when checking images, went to sile and replaced video digitizer card and confirmed
|proper operation. Will bench test old card to confimm issue.
Tuesday, November 02, 2010 Almeda Exit ALL Power issue caused fiber converters to be lurned off, Wailed on sile for power lo be restored and confirmed
proper operation of all equipment.
Tuesday, November 02, 2010 NEBWS Work John Ralston  |VESC configuration/installation
Tuesday, November 02, 2010 ALL ALL Completed aftemoon image checks of all lanes
Tuesday, November 02, 2010 Ship Channel Bridge ELPAC3 Issues relaled to comm during image checks, verified newest comm driver was installed and restarted
system. Verified proper operation.
Tuesday, November 02, 2010 Cullen Exit ALL Issues related to comm during image checks, verified newest comm driver was installed and restarted
system. Verified proper operalion.
Tuesday, November 02, 2010 1-10 TP1 NRT2 This system was not accessible remotely, rebooted with netbooter and confirmed proper operation.
Wednesday, November 03, 2010 ALL ALL Reviewed ATTLAS image report
Wednesday, November 03, 2010 ALL ALL Completed moming image checks of all lanes
Wednesday, November 03, 2010 ALL ALL Continued review of bad OCR data
Wednesday, November 03, 2010 Ship Channel Bridge ELPAC3 Issues related to comms during image checks again. Went to site and replaced comm board and confirmed
roper operation.
Wednesday, November 03, 2010 Almeda Exit ALL Power issue again, found bad comm card. Replaced comm card and confirmed proper operation.
Wednesday, November 03, 2010 NEBWS Work John Ralston JVESC configuration/installation
Wednesday, November 03, 2010 ALL ALL Completed afternoon image checks of all lanes
Thursday, November 04, 2010 ALL ALL Reviewed ATTLAS image report
Thursday, November D4, 2010 ALL ALL Completed moming Image checks of all lanes
Thursday, November 04, 2010 ALL ALL Continued review of bad OCR dala
Thursday, November 04, 2010 NEBWS Work John Ralston  JVESC configuration/finsiallation
Thursday, November 04, 2010 ALL ALL Completed aftemoon image checks of all lanes
Friday, November 05, 2010 ALL ALL Reviewed ATTLAS image report
Friday, November 05, 2010 ALL ALL Completed moming image checks of all lanes
Friday, November 05, 2010 ALL ALL Continued review of bad OCR data
Friday, November 05, 2010 I-10 TP3 NRT1 Unable to connect to this NRT remotely, rebooted using netbooter and verified proper operation.
Friday, November 05, 2010 ALL ALL |Worked on bench testing removed VES computers/LEDs and preparing them fo retum to service
Saturday, November 06, 2010 ALL ALL |Reviewed ATTLAS image report
Sunday, November 07, 2010 ALL ALL Reviewed ATTLAS image report
Monday, November 08, 2010 ALL ALL Reviewed ATTLAS image report
Monday, November 08, 2010 ALL ALL Completed moming image checks of all lanes
Monday, November 08, 2010 ALL ALL |Conlinued review of bad OCR data
Monday, November 08, 2010 Sam Central NRT |Unable to connect to this NRT remotely, rebooted using netbooter and verified proper operation.
Monday, November 08, 2010 WPT Hwy 6 ALL Issue relaled to comms during Image checks., Confirmed latest comm driver was installed, reslaried system
and confirmed proper operalion,
Monday, November 08, 2010 WestPark Extension ALL |Checked and reported incandecent bulb outages
Monday, November 08, 2010 Forl Bend Parkway ALL Checked and reported incandecent bulb outages
Monday, November 08, 2010 NEBWS Work Wilson Rd. VESC configuration/installation
Monday, November 08, 2010 ALL ALL Completed afternoon image checks of all lanes
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Tuesday, November 09, 2010 ALL ALL Reviewed ATTLAS image report
Tuesday, November 09, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 09, 2010 ALL ALL Continued review of bad OCR data
Tuesday, November 09, 2010 Telephone Rd Exit ALL lssue relaled to comms during image checks. Confirmed latest comm driver was installed, restarted system,
same issue. Went to site and replaced comm card. sama issue. Issue appears lo be software related.
Replaced system and confirmed proper operation, will re-image removed system.
Tuesday, November 08, 2010 NEBWS Work Wilson Rd. __|VESC configurationfinslallation
Tuesday, November 09, 2010 ALL ALL Completed afternoon image checks of all lanes
Wednesday, November 10, 2010 ALL ALL |Reviewed ATTLAS image report
Wednesday, November 10, 2010 ALL ALL Completed moming image checks of all lanes
Wednesday, November 10, 2010 ALL ALL Caontinued review of bad OCR data
Wednesday, November 10, 2010 NEBWS Work Wilson Rd. _|VESC configuration/installation
Wednesday, November 10, 2010 ALL ALL Completed afiernoon image checks of all lanes
Thursday, November 11, 2010 ALL ALL Reviewed ATTLAS image report
Thursday, November 11, 2010 ALL ALL Completed moming image checks of all lanes
Thursday, November 11, 2010 ALL ALL Continued review of bad OCR dala
Thursday, November 11, 2010 Lake Olympia ALL {Previously replaced VESC at this location with ELPAC, re-wirad to send RS422 comms direclly from the lan
controfler 1o the ELPAC and eliminate the RS232 to 422 convertor.
Thursday, November 11, 2010 ALL ALL Completed afternoon image checks of all lanes
Friday, November 12, 2010 ALL ALL Reviewed ATTLAS image report
Friday, November 12, 2010 ALL ALL |Completed moming image checks of all lanes
Friday, November 12, 2010 ALL ALL [Continued review of bad OCR data
Friday, November 12, 2010 ALL ALL IWorked on bench testing removed VES computers/LEDs and preparing them to retum to service
Friday, November 12, 2010 ALL ALL |Comgletad afternoon image checks of all lanes
Salurday, November 13, 2010 ALL ALL Reviewed ATTLAS image report
Sunday, November 14, 2010 ALL ALL |Reviewad ATTLAS image report
Monday, November 15, 2010 ALL ALL |Reviewed ATTLAS image report
Monday, November 15, 2010 ALL ALL Completed moming image checks of all lanes
Monday, November 15,2010 ALL ALL Continued review of bad OCR dala_
Monday, November 15, 2010 Cullen Exit ALL No triggers during moming image checks, LCA says lane controller was not working and they were in
rocess of replacing it.
|__Monday, November 15, 2010 1-10 TP2 NRT Unable to connect to this NRT remately, rebooted using netbooter and verified proper operation.
Monday, Noavember 15, 2010 NEBWS Work Wast Leke VESC configuration/finsiallation
Houston
Monday, November 15, 2010 ALL ALL Completed afterncon image checks of all lanes
Tuesday, November 16, 2010 ALL ALL |Reviewed ATTLAS image report
Tuesday, November 16, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 16, 2010 ALL ALL Continued review of bad OCR dala
Tuesday, November 16, 2010 Telephone Rd Exit ALL Issue related to comms during image checks. Confirmed latesi comm driver was Installed, restarted system
and confimed proper operation.
Tuesday, November 16, 2010 NEBWS Waork Wesl Lake VESC configurationfinstallation
Houston
Tuesday, November 16, 2010 ALL ALL |Completed afiernoon image checks of all lanes
Waednesday, November 17, 2010 ALL ALL |Reviewed ATTLAS image report
Wednesday, November 17, 2010 ALL ALL Completed moming image checks of all lanes
Wednesday, November 17, 2010 ALL ALL Continued review of bad OCR dala
Wednesday, November 17, 2010 Telephone Rd Exit Issue related to comms during image checks, Went to site and changed comm ports on the ELPAC and
confirmed proper operation
Wednesday, November 17, 2010 Airport Connector NRT NRT pre-production test passes
Wednesday, November 17, 2010 Little York Ent NRT |NRT pre-production test p
Wednesday, November 17, 2010 Litlle York Exit NRT |NRT pre-production test passes
Wednesday, November 17, 2010 Aldine Mail Route Exit NRT NRT pre-production test passes
Wednesday, November 17, 2010 NEBWS8 Work Wesl Lake VESC configuration/installation
Houston
Wednesday, November 17, 2010 ALL ALL Compleled aflemoon image checks of all lanes
Thursday, November 18, 2010 ALL ALL IReviewed ATTLAS image report
Thursday, November 18, 2010 ALL ALL |Completed moming image checks of all lanes
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Thursday, November 18, 2010 ALL ALL IContinued review of bad OCR data
Thursday, Noavember 18, 2010 Moore West ALL No triggers during morning image ched(s LCA says the issue is related lo loops and HCTRA will address
the issue
Thursday, November 18, 2010 ALL ALL Compleled aflemoon image checks of all lanes
Friday, November 19, 2010 ALL ALL |Revi d ATTLAS image report
Friday, November 19, 2010 ALL ALL ICompleted morming image checks of all lanes
Friday, November 19, 2010 ALL ALL [Conlinued review of bad OCR data
Friday, November 19, 2010 Airport Conneclor ALL Issues with video qualily during moming image checks, went to sile and replaced ground isolation
transformers and confirmed proper operation.
Friday, November 19, 2010 Hardy Tidwell Ent ALL Issues with video quality during moming image checks, went 1o site and replaced ground isolation
transformers and confirmed proper operalion.
Friday, November 19, 2010 ALL ALL Completed afternoon image checks of all lanes
Saturday, November 20, 2010 ALL ALL Reviewed ATTLAS image report
Sunday, November 21, 2010 ALL ALL Reviewed ATTLAS image reporl
Monday, November 22, 2010 ALL ALL Reviewed ATTLAS image repor
Monday, November 22, 2010 ALL ALL Completed moming image checks of all lanes
Monday, November 22, 2010 ALL ALL Continued review of bad OCR data
Monday, November 22, 2010 ALL ALL Installed windows XP patch on specific list of systems as requested by HCTRAIT team
Monday, No ber 22 2010 NEBWS Work Gamrett Rd VESC configurationfinstaflation
Monday, November 22, 2010 ALL ALL Completed afternoon image checks of all lanes
Tuesday, November 23, 2010 ALL ALL Reviewed ATTLAS image report
Tuesday, November 23, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 23, 2010 ALL ALL Continued review of bad OCR data dala__
Tuesday, November 23, 2010 NEBWS Work Garrelt Rd. Help with testing per HCTRA request
Monday, November 22, 2010 Red Bluff Exit ALL Issue related to comms during image chacks, Confirmed latest comm driver was installed, reslarted system,
confirmed proper operalion
Monday, November 22, 2010 Spencer Hwy Exit ALL Issue related to comms during image checks. Confirmed latest comm driver was installed, restarted system,
confirned proper operalion
Monday, November 22, 2010 Beltway 8 Exchange ALL - |!ssue related to comms during image checks. Confirmed latest comm driver was installed, restarted system,
confirmed proper operation
Tuesday, November 23, 2010 ALL ALL Completed aftermoon image checks of all lanes
Wednesday, November 24, 2010 ALL ALL Reviewed ATTLAS image repor
Wednesday, November 24, 2010 ALL ALL |Completed moming image checks of all lanes
Wednesday, November 24, 2010 ALL ALL Continued review of bad OCR data
Wednesday, November 24, 2010 NEBWS Work Wilson Rd. Help with testing per HCTRA ra@est
Wednesday, November 24, 2010 NEBWS Work Garett Rd. VESC configurationfinstallation
Thursday, November 25, 2010 ALL ALL Holiday
Friday, November 26, 2010 ALL ALL Holiday
Saturday, November 27, 2010 ALL ALL Reviewed ATTLAS image report
Salurday, November 27, 2010 Sam North ALL Alerts from NRT systems being down and not getting triggers. Unable to connect to some NRTs. Wenl lo
site and found breaker on UPS was tripped, reset breaker and all equipment recovered.
Sunday, November 28, 2010 ALL ALL Reviewed ATTLAS image report
Monday, November 29, 2010 ALL ALL Reviewed ATTLAS image report
Monday, November 29, 2010 ALL ALL Completed moming image checks of all lanes
Monday, November 29 2010 ALL ALL Continued review of bad OCR dala
Monday, November 29, 2010 ALL ALL Worked on bench tesling removed VES computers/LEDs and prepanng them lo retum to service
Monday, November 28, 2010 NEBW8S Work Winfield VESC configuration/installation
Monday, November 29, 2010 ALL ALL Completed afternoon image checks of all lanes
Tuesday, Novemnber 30, 2010 ALL ALL Reviewed ATTLAS image report
Tuesday, November 30, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 30, 2010 ALL ALL Continued review of bad OCR data
Tuesday, November 30, 2010 ALL ALL Worked on bench testing removed VES computers/LEDs and preparing them o retum lo service
Tuesday, November 30, 2010 NEBWS Work Winfield VESC configuration/installation
Tuesday, November 30, 2010 ALL ALL Completed aflernoon image checks of all lanes
Tuesday, November 30, 2010 ALL ALL

Work with HCTRA to complete VESC user password conversion and check every VES related system to
ensure they are working properly
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Tuesday, November 30, 2010 WPT Royallon Ent ALL ISystam did not come back up after password conversion, went onsite to reboot. Confirmed proper operation

Tuesday, November 30, 2010 Airport Conneclor ALL Comm issues with this system afier password conversion, went to sile to resolve issues, confirmed proper
operation.

Tuesday, November 30, 2010 Sam East PIC System did not come back up after password conversion, went onsite lo rebool. Confirmed proper operation
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LED Illummator

00557M 5

Repaired Equipment

B ot v Rroble M RGO AR IR | PFNE

REPLACED CONTROLLER BOARD AND 5 LED BANKS AND TESTED

4 LED BANKS NOT WORKING REPLACED 4 BAN KS AND TESTED

LED llluminator [01034M CONTROLLER BOARD BURNED REPLACED CONTROLLER BOARD AND TESTED
ILED llluminator j00504M CONTROLLER BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED

LED llluminator |00910M ISSUE SWITCHING TO DAY NIGHT MODE |RESET CONTROLLER BOARD, REPROGRAMMED AND TESTED
LED llluminator {00616M CONTROLLER BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED

LED llluminator {0201 CONTROLLER BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED

LED flluminator |00421M CONTROLLER BOARD BURNED REPLACED CONTROLLER BOARD AND TESTED

LED llluminator |01047M 1 LED BANK NOT WORKING REPLACED 1 BANK AND TESTED

LED llluminator [0353M 5 LED BANKS OUT — ED ( ER

LED llluminator {00389M CONTROLLER BOARD HAS NO POWER _|REPLACED CONTROLLER BOARD AND TESTED
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Invoice Mumber; 10-07310

Bilt To:
HARRIS COUNTY TOLL RD AUTHORITY
ACCTOUNTS PAYABLE
330 MEADOWFERN, SUITE 200
HOUSTON, TX 77057

Zusiomer Numuer HO013

Zustomer PO Number: @gﬁ/"}f | 5 {_5,? q fLP
1100007

Project Number;
Ptepanl Momge
Jerms,

Due Dale

NET 30
12/30/2010

Billing Pariod From: 11/01/2010
To: 1143012010

Current
Amount ]
5 VES Maint Techs 58,333.33
BiLL SCHEDULE ITEMS 58,333.23
mvoice Teial 58,333.23

. —00,000 = 14= T&333:33
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HCTRA Monthly VES Maintenance Report NOV10

g
TR

= Date - ..

= . Location =7 i
0 T e

J st Lane o Y.

o e S i e TR I AR L SemcePnflormed T = & L N P

- Ongmal Equipment Install Date +

Monday, November 01, 510 ALL ALL Ruvnwed ATTLAS image report =
Monday, November 01, 2010 ALL ALL Completed marning image checks of all lanes

Monday, November 01, 2010 ALL ALL Continued review of bad OCR dala

Monday, Novemher 01, 2010 ALL ALL Worked an bench testing removed VES computers/LEDs and preparing them lo refum to service

Monday, November 01, 2010 Airpan Conneclor ALL

Intermittant comm ssue on some of the lanes, went and swapped oclopus cable and confimed proper
oparalion

Monday, November 01, 2010 NEBWS Work John Ralston  |VESC configuration/installation
Monday, November 01, 2010 ALL ALL Compleled 1 image checks of all lanes
Tuesday, November 02 2010 ALL ALL Reviewed ATTLAS image repon
Tuesday, November 02, 2010 ALL ALL " Completed moming image checks of all lanes A
Tuesday: November 02, 2010, ALL T ALL Coniinued review ol bad OCR dala i '
Tuesday, November 02, 2010 Fallbrook Exit ALL Video qually issue when checking images, wen! (o sie and replaced video digitizer card and confirmed
. kwooet operation. Will bench 1es! old card lo confirm issue.
Tuesday, November 02; 2010 Almeda Exi ALL Power issue caused fiber conveners 1o be turned off. Wailed on site for power (o be restored and confirmed
roper op 1 of all equip
Tuesday, November 022010 NEBWS Waork John Ralston _JVESC configuration/instatiation
Tuesday, November 02, 2010 ALL ALL Completed afternoon image checks of all lanes
Tuesday, November 02, 2010 Ship Channel Bridge ELPAC3I | |Issues related to comm during image chech.s, venfied newesl comm dnvar was installed and resiarted
sysiem. Verfied proper operation.
Tuesday, November 02, 2010 Cullen Exit ALL Issues related 1o comm during image checks, verified newest comm driver was inslalled and reslarad
system. Verihed proper operation,
Tuesday, November 02, 2010 1-10 TP1 NRT2 This syslem was nol accessible remolely, rebooted with netbooter and confiumed proper ooaration.
Wednesday. November 03, 2010 ALL ALL Reviewed ATTLAS image report
Wednesday, November 03. 2010 ALL ALL Compl ing imaoe checks of all lanes
Wednesday. Navamber 03, 2010 ALL ALL Continued review of bad OCR data
Weunesday, November 03, 2010 Ship Channel Bndge ELPAC3 Issues related to comms dunng image checks again, Went (o site and replaced comm board and confimed
proper opetation.
Wednesday. November 03, 2010 Almeda Exit ALL Power issue again, found bad comm card. Replaced comm card and confimed proper operation.
Wednesday, November 03, 2010 NEBWE Work John Ralstion  |VESC configuralionfinsialiation
Wednesday, November 03, 2010 ALL ALL Completed afternoon image checks of all lanes
Thursday, November 04, 2010 ALL ALL Reviewed ATTLAS image repart
Thursday, Navember 04, 2010 ALL ALL Completed moiming image checks of all lanes
Thutsday, November 04, 2010 ALL ALL Continuad review of bad OCR dala
Thursday, November 04. 2010 NEBWS Work John Ralston |VESC confiauration/installation
Thursday, November 04, 2010 ALL ALL Compleled afternoon image checks of all lanes
Friday, Hovember 05, 2010 ALL ALL |Reviewed ATTLAS image report
Friday, November 05, 2010 ALL ALL Completed morning image checks of all lanes
Friday, November 05, 2010 ALL ALL Continued review ol bad OCR data
Friday. November 05, 2010 1-10 TP3 NRT1 Unable to connect to this NRT remotely. rebooied using netbooler and verified proper operation.
Friday, November 05, 2010 ALL ALL |Worked on bench testing removed VES compulers/LEDs and preparing them lo retum to service
Salrday. November 06, 2010 ALL ALL Reviawed ATTLAS image repon
Sunday, November 07, 2010 ALL ALL Reviewed ATTLAS image report
Manday, November 08 2010 ALL ALL Reviewcd ATTLAS image re
Monday, November 08, 2010 ALL ALL Completed moming image checks of all lanes
Monday. November 08, 2010 ALL ALL Conlinued review ol bad OCR dala
Monday. November 08, 2010 Sam Central NRT Unable 10 connect to this NRT remolely. rebooted using neibooter and venfied proper ogeration.
Monday, November 08, 2010 WPT Hwy & ALL Issue related 'o comms during image checks. Confirmed lalest comm driver was installed, resianed sysltem
Jand confimed proper oparation.
Monday. November 08, 2010 WaestPark Extension ALL Checked and reponed incandecent bulb oulages .
Monday, November 08, 2010 Fort Bend Parkway ALL Cheched and repored incandecent bulb outages
Monday, November 08, 2010 NEBWE Work Wilsan Rd. VESC configuration/inslallation
Monday, November 08, 2010 ALL ALL Compleled ahernoon image checks of all lanes
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Tuesday. November 09. 2010°"" T T ALL ALL Reviewed ATTLAS imace report
Tuesday, November 09, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 09, 2010 ALL ALL Continued review of bad OCR data
Tuesday, November 09, 2010 Telephone Rd Exit ALL Issue related to comms during image checks. Confirmed lalest comm driver was inslalied, restarted system,
same ssue. Wenl lo sile and replaced comm card, same issue. Issue appears 1o be software relaled,
Replaced system and confirmed proper operation. will re-image removed system.
Tuesday, November 09, 2010 NEBWS Work Wilson Rd. VESC canfigurationfinstallation
Tuesday, November 09. 2010 ALL ALL Compleled aflernoon image checks of all lanes
Wednesday. November 10, 2010 ALL ALL |Reviewed ATTLAS image repor
Wednesday. November 10, 2010 ALL ALL Completed moming image checks of all lanes
Wednesday. November 10, 2010 ALL ALL Continued review of bad OCR data_
Wednesday, November 10, 2010 NEBWS Work Wilson Rd. VESC cnni!guratnn.ﬂnsialatiun
Wednesday. November 102010 ALL ALL Completed afiernoon image checks of all lanes B
Thursday, November 11. 2010 : ALL 5 ALL Reviewed ATTLAS image repont ‘
Thursday. November 11. 2010 . : ALL ALL Completed moming image checks of all lanas
Thursday, November 11, 2010 ALL ALL Continued review of bad OCR dala
Thursday, November 11, 2010 Lake Olympia ALL Previously replaced VESC al this locaiion with ELPAC, re-wwred 10 send R5422 comms direcily from the lane]
: : . - controller to the ELPAC and eliminate the RS232 10 422 convertor.
Thursday, November 11, 2010 ALL ALL Completed afternoon image checks of all lanes
Frday, November 12, 2010 ALL ALL |Reviewed ATTLAS image repont
Friday. November 12, 2010 ALL ALL Compleled moming image checks of all lanes
Friday, November 12, 2010 ALL ALL Continued review of bad OCR data
Friday. November 12, 2010 ALL ALL Warked an bench testing removed VES computers/LEDs and prepaning them lo relum 1o service
Fnday. N ber 12, 2010 ALL ALL Completed aemoon image checks of all lanes -
Saturday, November 13, 2010 ALL * ALL Reviewed ATTLAS image report p
Sunday. November 14, 2010 ALL - ALL Reviewed ATTLAS image report X
Monday, November 15. 2010 ALL ALL Reviewed ATTLAS image repon £
Monday, November 15, 2010 ALL ALL Compieted maming image checks of all lanes
Monday. Novernber 15, 2010 ALL ALL Conlinued review of bad OCR data__- b
Monday, November 15, 2010 Culien Ext ALL No triggers during morming image checks, LCA says lane controller was not working and they were in
process of replacing il
Monday. November 15, 2010 1-10 TP2 NRT Unable 1o connect lo this NRT remalely, rebooled using netbooter and venfied proper of
l4onday, November 15, 2010 NEBWS Work Waes! Lake VESC configurationfinstaliation
Houstan
tonday. November 15, 2010 ALL ALL * |Completed atiernoon image checks of all lanes
Tuesday, November 16, 2010 ALL ALL Reviewed ATTLAS image report
Tuesday, No ber 16. 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 15, 2010 ALL ALL Continued review of bad OCR data
Tuesday, November 16, 2010 Telephone Rd Exi ALL Issue related to comms during image checks, Confirmed lalest comm driver was installed, restarted sy
and confimed praper operation.
Tuesday, November 16, 2010 NEBWS Work Wesi Lake VESC configuration/installation
Houston
Tuesday, N ber 16, 2010 ALL ALL Completed afiernoon image chechs of all lanes
Wednesday. November 17, 2010 ALL ALL Reviewed ATTLAS image report
Wednesday, November 17, 2010 ALL ALL Completed moming image checks of all lanes
Wednesday, November 17, 2010 ALL ALL Continued review of bad OCR dala
Wednesday, November 17, 2010 Telephona Rd Exit Issue related 1o comms during image checks. Went o sile and changed comm pors on the ELPAC and
i_mnﬁmed | proper operation
Wednesday November 17, 2010 Airport Connector NRT NRT pre-produchon lesl passes
| Wednesday, November 17. 2010 Litile York Ent NRT INRT pre-production tes! passes
Wednesday, Novemnber 17, 2010 Little York Exi NRT NRT pre-production test passes
Wednesday, November 17, 2010 Aldine Mail Route Exit NRT NRT pre-produclion les! passes
Wednesday, November 17, 2010 NEBWS Work West Lake | |VESC configuration/installation
Houslon
Wednesday, November 17, 2010 ALL ALL Completed afternoon image checks of all lanes
Thursday. November 18, 2010 ALL ALL Reviewed ATTLAS image repon z
Thursday, November 18, 2010 ALL ALL Completed moming image checks of all lanes
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- Thursday, November 18. 2010 ALL “ALL Continued review of bad OCR data e B
Thursday, November 18, 2010 Moore Wesl ALL No nggers duning morning image checks, LCA says the issue is related 1o loops and HCTRA will address
the issue
Thursday, November 18, 2010 ALL ALL Completed afternoon image checks of ail lanes
Frday. November 19, 2010 ALL ALL |[Reviewed ATTLAS image report
Friday, November 19. 2010 ALL ALL . Completed moming image checks of all lanes
Frday, November 19, 2010 ALL ALL Conunued review of bad OCR dala
Friday, November 19, 2010 Airpont Connectar ALL |Issues with video qualty dunng morming image checks, wen! to sile and replaced ground isclation
transformers and confirmed proper operation.
Fnday, November 19, 2010 Hardy Tidwell Ent ALL Issues with video quably dunng morning image checks, wenl Lo sie and replaced ground isolation
translormers and confirmed proper operalion.
Friday. November 15, 2010 ALL ALL |Compleled atiemoon image checks of all lanes
Saturday, November 20, 2010 ALL ALL Reviewed ATTLAS image repor
Sunday,.November 21, 2010 ALL TALL - Reviewed ATTLAS image repon :
Monday. Hovember 22, 2010 ALL ALL JReviesed ATTUAS image repon '
Monday, November 22, 2010 ALL ALL Completed moming image checks of all lanes
. Monday, November 22, 2010 ALL ALL Continued review ol bad OCR data
Mondayv. November 22, 2010 ALL ALL - Installed windows XP palch on specific list of systems as requestes by HCTRA IT team
Monday, November 22. 2010 NEBW8S Work Garrett Rd. VESC configurationfinstallation
Monday. November 22, 2010 ALL ALL Completed afternoon image chechks of all fanes
Tuesday. November 23. 2010 ALL ALL Reviewed ATTLAS image report
Tuesday. November 23, 2010 ALL ALL Completed moming image checks of all lanes
Tuesday, November 23, 2010 ALL ALL Continued review of bad OCR dala
Tuesday, November 23. 2010 NEBWE Work Garrett Rd. Help with 1estin t HCTRA reques!
Monday, November 22, 2010 Red Bluff Exit ALL Issua related to comms during image checks. Confirmed latest comm driver was insialled, restanted system,
g confirmed proper operation
Monday, November 22, 2010 Spencer Hwy Exit ALL Issue related to comms during image checks. Confirmed latest comm driver was inslalled d sysiem,
confirmed proper operation
Maonday, November 22, 2010 Beltway B Exchange ALL .|issue relaled to comms during image checks. Confirmed laiast comm driver was installed, restaried system,
Jconfirmed proper operation
Tuesday, November 23, 2010 ALL ALL Compleled afiernoon image checks of all lanes
Wednesday, Navember 24, 2010 ALL ALL Reviewed ATTLAS image report
Wednesday November 24, 2010 ALL ALL Complated moming image checks of all lanes
Wednesday, November 24, 2010 ALL ALL Conlinued review of bad OCR data
Wednesday, November 24 2010 NEBWS Work Wilson Rd. Help with testing per HCTRA request
Wednesday, November 24, 2010 NEBWS Work Garrett Rd. VESC configurationfinstaliation
Thursday, November 25, 2010 ALL ALL Holiday
Friday. November 26, 2010 ALL ALL Holiday
Salurday. November 27, 2010 ALL ALL Reviewed ATTLAS image repornt
Saturday, November 27, 2010 Sam North ALL Alerts from NRT sysiems being down and not geilling inggers, Unable to connect to some NRTs. Wenl 1o
site and lound breaker on UPS was inpped. resel breaker and all equipment recovered
Sunday, November 28, 2010 ALL ALL Reviewed ATTLAS image report
Monday, November 25, 2010 ALL ALL Reviewed ATTLAS image repor
Monday. November 29. 2010 ALL ALL Completed ing image checks of all lanes
Monday, November 29, 2010 ALL ALL Continued review of bad OCR data
Monday. November 25. 2010 ALL ALL Worked on bench lesting removed VES computers/LEDs and prepanng them 1o relum 10 service
Monday. November 29, 2010 NEBWWE Work Winfield VESC configurationfinstallation
Monday, November 29, 2010 ALL ALL Compleled afiernoon image checks of all lanes
Tuesday. November 30. 2010 ALL ALL Ruviewnd ATTLAS image report
Tuesday, November 30. 2010 ALL ALL Completed moming image checks of all lanes
Tuesday November 30, 2010 ALL ALL Continued review of bad OCR dala
Tuesday, November 30. 2010 ALL ALL Worked on bench lesting removed VES compulers/LEDs and preparing Inem 1o retum to service
Tuesday. November 30, 2010 NEBWS Wark Winfield VESC confiquration/ Hali
Tuesday, November 30, 2010 ALL ALL C | anernoon image checks of all lanes
Tuesday: November 30, 2010 ALL ALL

Hsnsure iney are working propery

Work. wath HCTRA 10 complete VESC user password conversion and check. every VES related sysiem 10
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System did nol coma back up alter password conversion, went onsite lo reboot. Confemed proper operation

Tuesday, November 30,.2010 . WPT Royahaon Enl ALL

Tuesday, November 30, 2010 Aiport Connector ALL Comm issues with this sysiem after password conversion, wenl 10 sie to resolve issues, confirmed proper
operation. '

Tuesday, November 30, 2010 Sam East PIC System did not come back up afier password conversion, went onsite (0 reboot. Confirmed proper operauon
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Repaired Equipment

|-.Equipment«

.Serial Number

oy . 2SN Problem 30 e L

ELTEY

W e R LResolutiont™ wtas - 0 el T

{LED llluminator

00557M

4 LED BANKS NOT WORKING

REPLACED 4 BANKS AND TESTED

ILED Illuminator

01034M

CONTROLLER BOARD BURNED

REPLACED CONTROLLER BOARD AND TESTED

r

LED llluminator |00504M CONTROLLER BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED
LED llluminator |00S10M ISSUE SWITCHING TO-DAY NIGHT MODE |[RESET CONTROLLER BOARD, REPROGRAMMED AND TESTED
-~ ILED:Hlluminator {00616M CONTROLLEER BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED
LED llluminator 0201 CONTROLLER'BOARD HAS NO POWER |REPLACED CONTROLLER BOARD AND TESTED
. |LED-lluminator |00421M CONTROLLER BOARD BURNED REPLACED CONTROLLER BOARD AND TESTED
JLED llluminator [01047M 1 LED BANK NOT WORKING REPLACED 1 BANK AND TESTED
|LED llluminator {0353M 5 LED BANKS OUT REPLACED CONTROLLER BOARD AND 5 LED BANKS AND TESTED
ILED lliuminator |00389M CONTROLLER BOARD HAS NO POWER

REPLACED CONTROLLER BOARD AND TESTED
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Haris County Production

PO Reference Report

DRGANIZA TION NAMIE

PURCHASE DRIER S0

H/C TOLL ROAD AUTHORITY P149763
VERINIK MO Iate REQUEST NO
251730334 05/20/2010 R283707
NUYER Namil EST DELIVERY DATE
.o VGROCE
VENDOR ADDRESS: - sHIpTo: BILL TO:
TRANSCORE © 1 HCTOLL AUTHORITY
2705 WEST SAM HOUSTON PKWY 1 * CENTRAL SUPPLY
HOUSTON, TX 77043 . 14640 HENRY ROAD
: - HOUSTON, TX 770603623
CONTACT:
713 355 3628
ln,\mn OTY_ _UNIT _ DESCRIFTION © : o UNITPRICE .
[RLY] 750 000 Uﬂ 1

“PURCHASE ﬂiil)! R l'\' I U\D :?Iﬂ I OR 6
MIN‘I HS OF PRIJECTED EXPENDITURES DUIE
THE UNAVAILABILITY OF COMMERICAL
PPAPER PROCEEDS 1IN FURTD 3950,

PPURCHASE ORDER P13537 WILL BE REDUCED
1Y $750.000.00.

Onm 4 73000000 LOT

0

VES CAMERAS & ASSOCIATED EQUIPMENT

FOR HCTRA, ALL IN ACCORDANCE WITH
AGREEMENT HERETO REFERRED AND MADE PART
HEREOF AN'TF ATTACHED,

COMMISSIONERS COURT AFPROVELD 1722708

0u02. 4,000,0.00 LOT ADDENDUM R2R85456-1, DATED 12429408,
TOADD FUNDS TO PURCHASE ORDER
PRA9763 IN FUND 3710 FOR 6 MONTHS OF
PROJECTED EXPENDITURES DUE TO THE
UNAVAILABILITY OF COMMERCIAL PAPER
PROCEEDS IN FUND 3930, PURCHASE ORDER
P134347 WILL BE REDUCED BY
S4.000 o0

VES CAMERAS & ASSOCIATED EQUIPNMIEENT
FOR HCTRA. ALL IN ACCORDANCE WITH
AGREEMENT HERETO REFERRELD AND NMADE
PART HEREOF AN IF ATTACIHED,

noo3 2.544,550.00 LOT ADDENDUM TO INCREASE 149763 BY
$2544,550.68 WHICH REPRESENTS THE
REMAINING BALANCE ON THE RELATED PO,
1M1 34547,

MATERIALS & LAROR: VIOLATION ENFORCEMENT

SYSTEM CAMERAS & ASSUCIATED EQUIPMENT

FOR IILIR:\ ALL IN ACCORDANCE WITH

INTHERETO REFERRED AND MADE PART

STF ATTACHED,

(IR T00,000.00 LOT ADDENDUNM TO INCREASLE 'I%9703 DY S700,000 FOR THE
SEGCOND \.!EJ\R RENEWAL OF THE ANNUAL MAINTENANCE

FIRST YEAR, 122/08-1/21500

- SECOND YEAR: 12200-1221/10

N3 TU0,000.00 LOT ADDENDUM RA23293, DATED /1300, T0O INCREASE
P1A9T63 BY STO0000 FOR THE RENEWAL OF
MAINTENANCE FOR VIOLATION ENFORECENMENT SYSTEM

[J\rr CCKIRKIAT - Kitkputrick, Connie Page: 1
"Epuﬂ PO_FORMAT_HC - Purchiase Order Report Format
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1.00 < 0040.000.00

1.00 2.304,550,00
1.00 700.000.00
1.0 T00.000.00

OVOZ2011 :Current Date
09:39:04 :Current Time



Kirkpatrick, Connie (Au di'tors)

From: Fischer, Felita (HCTRA) (Toll Road Authority)
Sent: Friday, January 07, 2011 2:40 PM

To: Cross, Sheila (HCTRA) (Toll Road Authority)
Cc: Kirkpatrick, Connie (Auditors)

Suhject: Transcore Invoice # 10-07310--558,333.33
Sheila/Connie,

Please pay Transcore invoice # 10-07310--$58,333.33 on PO # P156924.
Please let me know if you have any questions.
Thank you,

Felita Fischer
Maintenance
Harris County Toll Road Authority
7701 Wilshire Place Drive
Houston, Texas 770400

713-587-7913 Office *
281-875-1592 Fax

felita.fischer@hetra.ory

* PLEASE NOTE NEW TELEPHONE NUMBER
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RECEIVING FORM IRAAP AUGI2' 109 o

PLEASE SPILT PAY THIS INVOICE ON TWO

Date: 08/04/10 DIFFERENT PURCHASE ORDERS

Section/Person Responsible for PO: Facilities-Maintenance/Felita Fischer
P149763--$278,384.00 &

Purchase Order# P156924--$22,197.67 Approved -~

Vendor Name: TransCore Date £— x 3—ie

Invoice # : 19330 Batch AK 8/ 7 £/

Invoice Amount:  $300,581.67 CC Date

VES Controllers, Plaza Image Computers, Hardware, Software Licenses,
Mainline Lane Instaillation (Labor & Materials) and Lane Closures from

Description: May 13, 2010 through July 19, 2010
PI4IT63 ~623000-LINE ITEM #3 ON
ORG KEY: $/%% 283,04 OBJECT CODE: PO # P149763-$278,384.00
pist9at RTT5]E2A41T-LINE ITEM #ZON (o))
P.O. BALANCE:$7 733.3]).0/ OBJECT CODE: PO # P156924--$22,197.67 __,
i rd
Approved for Payment
Q_,OQ/{QZ(A.ZJ/\, & DJ/ O Sectiones\pproval
Authorized Signature Date Revised 10/2/06
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"ICAUG 18 ru12:47

b INVOICE

TRANS

GECOF

RE.

Invoice Ho, 18330

Ll oy

Vendor Mo, 251730334

TransCore Proiect Mo, 1308121

o

Invosca Oata: 7/22/10

Dua Dat; 2210

Customar Ho.: 110013

Hame of Projact; HCTRA VES UPGRADE
Nama 01 Conuocior: TransCore

Billta: Harnis County Toll Road Authority
7701 wilshire Place Driva, 5th tloor
Haustan, TX 77040
Anantion; Chuck Sanders

Raomii to:

TransCora Holdings, Inc.
36120 Troasury Contar
Chicago, Il 60694-6100

SUMMARY OF WORK FOR WHICH PAYMENT IS REGUIRED

‘ Tewl Pravious Curiunt
'| Contruct Uni Contruct Total Contract Cunnitity Pravious Amouni Monin Currant Month
: Hetttt hem Quantity Amount Amount Invoiced Invoiced Quanuty Amount
1 600 % 177800 § 100740000 603 S f07n7atoD, O s -
2 VES Crmaa 1ibusing we % 222000 § 80136000 330 % 601,360.00 U] 3 -
3 Comurn Junctan 1sas |1 esmerninna) 02 $ LIG000 & 9463,380.00 340 i 5 1,084 000,00 1 0 § b
4 Canma Juncron Bas 12 cummananag) 93 £ 364500 ¢ 240,885.00 65 g 00 0 tls 5
5 . LED Mumionior 500 201000 § 145050000 500 |$ 140,60060¢ O ) -
b, Ho Cturrge LD thaminatlos s T - |/ s -0 S -
G LED Ruminntor Besics el 500 < 13000 ¢ £5,000.00 500 , $ 65,000,001 1] § -
b, Ho Cheagn | ED Mamnator 17ch ot H - s 2 3 s -, 0 3 -
7 LED Muminator Junntion Bos 11 174300 % - 0 5 = o 5 -
i 8  VES Conroler (1-4 cammas) 40 H 835100 § 33416000 57 s NG00 0 i 1% .
) 5 VES Contrglher (1-8 ciimexas) 65 s 810100 £ 59156500 75 s nE2LIS00 8 > o & 54,606.00
g 10 Maza knage Compuier 1 S 887700 $ 8.677.00 0 3 A x § 6772800
11 8. OCH Sysiem Hirdwin 12 5 588400 % B82,728.00 ] | 5 P [:] £ 5 55.15200
o, OCR Sysiem Sonware License " s 108000 £ 44,880.00 0 H - 3 s 32,640.00
12 $ 3092600 § - '] s . 1] b .
13 inibal Commissioting aod Tasting 1] $ 1201002 ¢ - ] : - o 5 -
Sutnotal - Matarials 4 5.786,535.00 2608 3 6,016,725 04 b $ 210,137.00
14 ! {Laver « » ints} 0 s 961500 § - 7 5. 67,305 00 0 $ B
15  Qmer' Lane instillasen {1Labor » Milknals) 300 4 2274000 $ 887211000 72 : . 1,637,628 00 3 s £8,247.00
16 Singln Lanailamp Closue - TrensCora Wosk L] M7 § - H - it 1 15,648.22 )
7 Two-Lana Closwra - IrmnsCore Won, § “1.30689 % - tH - 5 5 0.54B,45
18 Singht Lk Itamp Closure - Otfiers Work § 177500 § - S - 0 $ o
16 Twolane Closurn - Othvrs Wors $ 212555 § - 1 - 1] - -
20 Public Hottication - Weas 5 67500 § - 5 ; - 1] ] -
Subiiotal - Labior i BABIZ.M0.00 ) 1,705,233.00 ¥ 90,444.67
Cumant imvoico Amosnt {3 30058167
— 2o . ;
K 7 . Total Amount invoiced 1o Date | § 8.022.539.67 |
MA fc‘o.u'u ]
Oiner: T122/2010 Controct Amount § 14,058.645.00
TransCorg Amnount Ramilning 3  6,636,105.33

Shier Pay od Tdo L0O.'s

D 149763 - “#37& 384 oo
Pistqay- 433 19747
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May/June/July Lane Closures

Date Location Type Quantity  Price Date Issue
) tnvoiced

5/13/2010|249 Enirance Lanes 31,32 Single Lane/Ramp Closure i $1,117.73 Camera in lane 31 image is tilted [Ci
to the left m
Vi
5/20/2010|Bellaire Exit Lanes 68,69 Singls Lane/Ramp Closure 1 $1,117.73 Swap FSJ4 to LDF4 per HCTRA |R
ct
-

5/25/2010{West Ra Evit Lanes 58,59 Single Lane/Ramp Closure 1 $1,117.73 Camera in lane 58 has dark C
images re
al

5/26/2010|Bellaire Entrance 60,61 Singl2 Lane/Ramp Closure i $1,117.73 Camera in Lane 61 has dark [
images 13

o

6/1/20101Sam Souih Lanes 7.8 Two-Lane Closure 1 $1,309.89 Center camera in lane 7 has dark{C
6/1/2010|Sam South Lane 8 Sinala Lans/Famp Closure 1 $1.117.73 images. Richt camerainlane 8 |s
6/10/2010|Sam Souihwesi Lane 1 Single Lane/Ramp Clasure 1 $1,117.73 On-Retro LED illuminator not €
working H]

Le]

6/10/2010|Moore Wes: Shoulder Lane Single Lane/Ramp Closure i $1,117.73 Damags caused by vehicle E
accident a

6/11/2010lMoore West Lanes 1.2 Two-Lane Closurs 1 51,308.89 Test all VES lichting and replace |F
6/11/2010|Moore West Lane 3 Sinale Larne/Esmp Closure i S1.117.73 liaht bulbs and sockets afier 3
6/1412010|Moore West Lanes 1,2 Two-Lane Closure 1 $1,309.89 Found bulbs out when powering [F
up entire system. £

6/16/2010|Sam Norh Lane 3 Single Lane/Ramp Closure 1 $1,117.73 Loop damage caused by vehicle [F
accident H

4

6/i7/2010|/Sam Norih Lane 6,7 Two-Lane Closure 1 $1,309.89 Center camera in lane 7 has dark |t
images. T

v

Ll

i il

6/23/2010|Bellaire Exit Lanes 68,69 Single Lane/Ramp Closure 1 $1,117.73 Camera in lane 68 has dark '
images 1

Sincle Lans/Ramp Closure 1 $1,117.73 Camera in Lane 70 was fuzzy !

6/22/2010

Rankin Enirance Lane 70

and out of iocus.
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6/2E/2010|Westheimer Enirance Lanes <0.41]Single Lane/Ramp Closure 1 $1,117.73 Install lane controt signal in lane |In
42 [

al

7/5/2010|Sam HMorih Lanes 5,6 Two-Lane Closure 1 $1,309.89 Cenier camera in lane 7 has dark|C
7/5/2010|Sam Norh Lane 7 Sincle Lane/Ramop Closure 1 $1,117.73 images €
7/12/201G|Sam South Lane & Single Lane/Ramp Closure 1 $1,117.73 Camera showing video out on R
ELPAC Gl

N
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